University of Waterloo STAT 220 – W. H. Cherry

#0.2a

[INSTRUCTOR’S VERSION – NOT TO BE GIVEN TO STUDENTS]

____________________________________________________ ____________________________________________________

STAT 220 VIDEO SCHEDULE: Fall Term, 2003

The following schedule is requested for showing 17 videos in the course STAT 220 in room RCH 306 on Mondays, Wednesdays and Fridays; classes run from 12.30 to 1.20 p.m. The room has 120 seats and about 55 students are currently enrolled in the course. Day Date Starting Video Title Lect. Length Time No. (min.)

Wed. Sep. 13 12.29 2 25

Against All Odds: Program 1 – WHAT IS STATISTICS? Wed. Sep. 20 12.29 6 25

Against All Odds: Program 2 – PICTURING DISTRIBUTIONS. Mon. Sep. 25 12.29 8 25

Against All Odds: Program 3 – DESCRIBING DISTRIBUTIONS:

NUMERICAL DESCRIPTION OF DISTRIBUTIONS. Fri. Sep. 29 12.29 11 25

Against All Odds: Program 4 – NORMAL DISTRIBUTIONS. Mon. Oct. 02 12.29 12 25

Against All Odds: Program 5 – NORMAL CALCULATIONS. Mon. Oct. 16 12.29 17 25

Against All Odds: Program 18 – THE SAMPLE MEAN AND CONTROL CHARTS.

Fri. Oct. 20 12.29 19 25

Against All Odds: Program 19 – CONFIDENCE INTERVALS. Fri. Oct. 27 12.29 22 25

Against All Odds: Program 15 – WHAT IS PROBABILITY? Mon. Nov. 06 12.29 26 25

Against All Odds: Program 16 – RANDOM VARIABLES. Fri. Nov. 10 12.29 28 25

Against All Odds: Program 17 – BINOMIAL DISTRIBUTIONS. Mon. Nov. 13 12.29 29 25

Against All Odds: Program 13 – BLOCKING AND SAMPLING:

EXPERIMENTS AND SAMPLES. Fri. Nov. 17 12.29 31 25

Against All Odds: Program 14 – SAMPLES AND SURVEYS:

SAMPLING AND SAMPLING DISTRIBUTIONS. Wed. Nov. 22 12.29 33 19

SAMPLING AND ESTIMATION. Fri. Nov. 24 12.29 34 25

Against All Odds: Program 8 – DESCRIBING RELATIONSHIPS. Wed. Nov. 29 12.29 36 25

Against All Odds: Program 9 – CORRELATION. Fri. Dec. 01 12.29 37 25

Against All Odds: Program 11 – THE QUESTION OF CAUSATION. Mon. Dec. 04 12.29 38 25

Against All Odds: Program 12 – EXPERIMENTAL DESIGN.